Author's response to reviews

Title: Evaluating severity of neonatal jaundice using a modified bilirubin-induced neurologic dysfunction (BIND-M) score in a resource-limited setting

Authors:

Paula G Radmacher Dr (pgradm01@exchange.louisville.edu)
Frank D Groves Dr (frank.groves@louisville.edu)
Bolajoko O Olusanya Dr (BOOLUSANYA@aol.com)
Joshua A Owa Prof (jaowa2001@yahoo.co.uk)
Gabriel E Ofowwe Prof (ofowwegabriel@yahoo.com)
Emmanuel A Amuabunos Dr (amua4@yahoo.com)
Tina M Slusher Dr (tslusher@umn.edu)

Version: 3
Date: 19 November 2014

Author's response to reviews:

In this revised version:
1. We have added line numbering to the main manuscript.
2. In authors' contributions we replaced the authors' surnames with their initials; and added that all the authors read and approved the final copy of the manuscript.
3. We have also added that the authors do not have any financial or other competing interests.